Author’s response to reviews

Title: Low birth weight contributed to increased serum IL-6 levels in infantile respiratory syncytial virus infection

Authors:

Yuanjian Sheng (shengyj1978@163.com)
Shanshan Xu (294723131@qq.com)
Xuejing Li (lixuejing200806@163.com)
Jinling Liu (horseliu@126.com)
Xiling Wu (wuxiling@zju.edu.cn)
Xuefeng XU (xuxuefeng@zju.edu.cn)

Version: 4 Date: 21 Oct 2017

Author’s response to reviews:

Dear editor:
We carefully studied reviewer’s comments again, and have made the necessary changes according to the suggestions. Our responses to reviewer’s comments are listed below:
1. Please note that the initials "SS" listed in the Authors' contributions does not correspond to any of the authors listed on the title page; please correct this discrepancy.
   Reply: The initials "SS" listed in the Authors' contributions does not correspond to the authors listed on the title page, “SS” had been deleted.

2. In the Funding section, please indicate the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.
   Reply: “The study sponsors had no involvement in the design of the study, collection, analysis, interpretation of data, and the writing of manuscript” had been added in the Funding section.

3. Please upload a clean version of the manuscript without tracked changes or highlighting and then remove any highlighted versions from the inventory, as these are no longer required at this stage of the editorial process.
   Reply: We had uploaded a clean version of the manuscript without tracked changes or highlighting. We also removed the highlighted versions from the inventory.

We have addressed all the suggestions of reviewer. Undoubtedly, this revision will improve the quality and clarity of this manuscript. We thank again you for your consideration of this manuscript.
Sincerely yours,
Dr. Xue-Feng Xu,
E-mail: xuxuefeng@zju.edu.cn
The Children's Hospital, Zhejiang University School of Medicine, Hangzhou 310052, P.R. China